CSCI 2321 (Principles of Computer Design), Spring 2002 Homework 1

Assigned: January 18, 2002.

Due: January 25, 2002, by the start of class. Credit: 10 points.

1 Problems

Write out answers to the following questions (by hand or using a word-processing or text-formatting program). Please submit hardcopy.

1. (7 points) Do problems 1.1 through 1.26 on pp. 45–46 of the textbook. 2. (3 points) Do problem 1.56 on p. 50 of the textbook. (The problem asks you to write “at least one paragraph” on each of several questions. One-sentence paragraphs are acceptable if that’s all you have to say.)